

Spatio-temporal Genetic Structuring of Leishmania major in Tunisia by Microsatellite Analysis

Myriam Harrabi, Jihène Bettaieb, Wissem Ghawar, Amine Toumi, Amor Zaatour, Rihab Yazidi, Sana Chaabane, Bilel Chalghaf, Mallorie Hide, Anne-Laure Bañuls, et al.

▶ To cite this version:

Myriam Harrabi, Jihène Bettaieb, Wissem Ghawar, Amine Toumi, Amor Zaatour, et al.. Spatiotemporal Genetic Structuring of Leishmania major in Tunisia by Microsatellite Analysis. PLoS Neglected Tropical Diseases, 2015, 9 (8), pp.e0004017. 10.1371/journal.pntd.0004017. hal-01358510

HAL Id: hal-01358510 https://hal.science/hal-01358510

Submitted on 26 May 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

RESEARCH ARTICLE

Spatio-temporal Genetic Structuring of *Leishmania major* in Tunisia by Microsatellite Analysis

Myriam Harrabi^{1,2}*, Jihène Bettaieb¹, Wissem Ghawar¹, Amine Toumi¹, Amor Zaâtour¹, Rihab Yazidi¹, Sana Chaâbane¹, Bilel Chalghaf¹, Mallorie Hide³, Anne-Laure Bañuls³, Afif Ben Salah¹

1 Institut Pasteur, Tunis, Tunisia, 2 Faculté des Sciences de Bizerte-Université de Carthage, Tunis, Tunisia,

3 UMR MIVEGEC (IRD 224-CNRS5290-Universités Montpellier 1 et 2), Centre IRD, Montpellier, France

* myriamharrabi@yahoo.fr

Abstract

In Tunisia, cases of zoonotic cutaneous leishmaniasis caused by Leishmania major are increasing and spreading from the south-west to new areas in the center. To improve the current knowledge on L. major evolution and population dynamics, we performed multilocus microsatellite typing of human isolates from Tunisian governorates where the disease is endemic (Gafsa, Kairouan and Sidi Bouzid governorates) and collected during two periods: 1991–1992 and 2008–2012. Analysis (F-statistics and Bayesian model-based approach) of the genotyping results of isolates collected in Sidi Bouzid in 1991-1992 and 2008-2012 shows that, over two decades, in the same area, Leishmania parasites evolved by generating genetically differentiated populations. The genetic patterns of 2008–2012 isolates from the three governorates indicate that L. major populations did not spread gradually from the south to the center of Tunisia, according to a geographical gradient, suggesting that human activities might be the source of the disease expansion. The genotype analysis also suggests previous (Bayesian model-based approach) and current (F-statistics) flows of genotypes between governorates and districts. Human activities as well as reservoir dynamics and the effects of environmental changes could explain how the disease progresses. This study provides new insights into the evolution and spread of L. major in Tunisia that might improve our understanding of the parasite flow between geographically and temporally distinct populations.

Author Summary

In Tunisia, zoonotic cutaneous leishmaniasis (ZCL) constitutes a significant public health problem. Since 1884, the Gafsa, Kairouan and Sidi Bouzid governorates are the most endemic areas of ZCL. This study used a multi-locus microsatellite typing approach to study the evolution and the population dynamics of *Leishmania major* in Tunisia. Within the same area, in twenty years, parasite populations evolved by producing a genetically



OPEN ACCESS

Citation: Harrabi M, Bettaieb J, Ghawar W, Toumi A, Zaâtour A, Yazidi R, et al. (2015) Spatio-temporal Genetic Structuring of *Leishmania major* in Tunisia by Microsatellite Analysis. PLoS Negl Trop Dis 9(8): e0004017. doi:10.1371/journal.pntd.0004017

Editor: Gabriele Schönian, Charité University Medicine Berlin, GERMANY

Received: February 25, 2015

Accepted: July 30, 2015

Published: August 24, 2015

Copyright: © 2015 Harrabi et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The present study is part of a research project funded by US-NIAID-NIH, Grant number 1P50AI074178-01. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

differentiated population, probably better adapted to the ecosystem. In agreement with the reported human cases of ZCL, the genetic data on samples from the three governorates shows that the disease did not spread according to a geographical gradient. Furthermore, *L. major* flows seem to still occur between governorates and neighboring districts. This study suggests that environmental changes, human activities and reservoir systems have influenced the spread and evolution of *L. major* populations. Our findings provide important knowledge on the epidemiology of *L. major* in Tunisia and might help understanding why the disease is still spreading from the south to the center, despite the control measures that have been put into place.

Introduction

In Tunisia, zoonotic cutaneous leishmaniasis (ZCL), also known as "Le Bouton de Gafsa" (the pimple of Gafsa), was first described in 1884 by Déperet and Boinet in the Gafsa governorate (south-west of Tunisia) [1]. ZCL represents a typical model of emerging and reemerging zoonosis [2]. ZCL can cause substantial morbidity because of the presence of chronic skin ulcers and the psychological effect of disfigurement [3]. No vaccine is available yet and the current treatments (mainly intra-lesion injections) are expensive and not easy to administer, particularly to children and patients with multiple lesions. For this reason, an international research partnership was launched in 1995 to focus on clinical trials of topical preparations, mainly paromomycin ointments, as new treatments of ZCL caused by Leishmania major [4,5,6]. Moreover, epidemiological studies have attempted to determine the spatial and temporal dynamics of ZCL epidemics to improve the prediction of their occurrence and consequently their control [7.8]. ZCL has been endemo-epidemic in the Gafsa region for many years, and in 1982 an epidemic was recorded in the Kairouan governorate for the first time [9]. Then, the disease spread to Sidi Bouzid, where it emerged as an epidemic in 1991 (see map of Tunisia in Fig 1 to localize these regions) [10,11]. Since then, the disease is maintained in these areas and has expanded also to other governorates in the center and south of Tunisia [12].

L. major is transmitted by the sand fly vector *Phlebotomus papatasi* [13] and rodents are the reservoir, including *Psammomys obesus* (fat sand rat) and *Meriones shawi* (Shaw's jird) [14,15]. The majority of *L. major* strains isolated in Tunisia belong to the MON-25 zymodeme [11,16,17]. It is largely recognized that the population structure of pathogens is influenced by different evolutionary factors, particularly during invasion of new ecosystems [18]. However, it is not known how their geographical distribution and temporal emergence affected the genetic structure and evolution of Tunisian *L. major* populations, mainly because multi-locus enzyme electrophoresis (MLEE), which has been widely used for phylogenetic typing of parasites, is not discriminative enough. On the other hand, microsatellite markers in combination with recent statistical methods represent a powerful tool for studying the population structure and monitoring the dynamics of these pathogens in time and space because of their high discriminating power and their presumed neutrality [19,20]. Nevertheless, no detailed study has been carried out on the population structure of *L. major* in Tunisia with these powerful markers.

Therefore, the aim of this study was to analyze the spatio-temporal organization of *L. major* in the three main endemic areas of ZCL in Tunisia (Gafsa, Kairouan and Sidi Bouzid governorates) by using a multi-locus microsatellite typing approach and population genetic statistical methods. Human isolates collected in the three governorates between 2008 and 2012 were used to assess the population dynamics of *L. major* according to the geographical distribution and the chronological emergence of leishmaniasis epidemics in human populations. Moreover,





Fig 1. Map of Tunisia and schematic illustration of gene flow between the study areas. (A1) Location of Tunisia, North Africa. (B1) Location of the Gafsa, Sidi Bouzid and Kairouan Governorates within Tunisia. (C1) Zoom of the area under study to show Chrarda and Mnara (Nasrallah delegation) (Mnara and Chrarda are shown in the map with red asterisks) in the Kairouan Governorate.

doi:10.1371/journal.pntd.0004017.g001

human isolates collected in Sidi Bouzid governorate in the 1991–1992 and 2008–2012 periods were compared to provide information on the evolution of *L. major* populations in the same area over twenty years.

Material and Methods

Ethics statement

Research included in this study was subject to ethical review by the International Review Board of the Pasteur Institute of Tunis and approved as part of a research project submitted in response to a "call" from the National Institutes of Health for funding Tropical Medicine Research Centers. The Pasteur Institute of Tunis, the study sponsor, took out an insurance policy in accordance with the Tunisian law concerning clinical research. All clinical or biological investigations were performed in the Health Centers of each governorate to guarantee the patients' safety, confidentiality and respect. All human samples of *Leishmania* were isolated from patients after collection of the written informed consent at the time of the clinical examination. Human isolates were anonymized and the associated information coded for privacy-preserving data mining.

Geographical origin and collection time of the biological samples

A total of 172 parasite samples, taken from the swollen edge of cutaneous lesions, were collected from patients in different governorates of Tunisia. Isolates from Sidi Bouzid governorate were collected during the 1991–1992 (historical isolates) and the 2008–2012 (recent isolates) periods, whereas isolates from Gafsa and Kairouan were only from the 2008–2012 period (recent isolates) (<u>Table 1</u>). Isolates were stored in the cryobank of the Department of Medical Epidemiology, Pasteur Institute of Tunis. Samples were isolated and typed in the framework of different research projects related to ZCL in Tunisia. Isolates were identified as *L. major* by sequence analysis of the gene encoding RNA polymerase II [21] in combination with the MLEE technique [22] at the Centre National de Référence des Leishmanioses (CNRL) of Montpellier, France.

Population dynamics were investigated at the: i) spatial scale, using recent isolates from Gafsa (GF2008-2012) (66 strains), Kairouan (KR2008-2012) (46 strains) and Sidi Bouzid (SBZ2008-2012) (25 strains) and ii) temporal scale, using the recent isolates (SBZ2008-2012, n = 25 samples) and the historical isolates (SBZ1991-1992, n = 35 samples) from the Sidi Bouzid governorate. Furthermore, the recent isolates from the Kairouan governorate (KR2008-2012, 46 strains) were subdivided in two groups based on the history of leishmaniasis infection in the area: 15 isolates from the historical focus of Mnara (MN2008-2012), where the first outbreak occurred in 1982, and 31 isolates from the recent focus of Chrarda (CH2008-2012), where the epidemic started only in 2005 (Fig 1 and Table 1).

DNA extraction and microsatellite genotyping

For all samples, total genomic DNA was extracted from parasite mass cultures (promastigotes) using DNA extraction kit, according to the manufacturer's protocol. DNA quality was checked by agarose gel electrophoresis and the concentration measured with a NanoDrop spectrophotometer. Amplification was carried out using fluorescent-labeled primers for ten polymorphic microsatellite markers and the PCR conditions previously described for *L. major* [20]. 1µl of PCR product was added to a standard loading mix with 0.5µl of internal lane size standard, and 13.5µl of formamide. Genotyping was performed by capillary electrophoresis using an

Table	1.	Geograp	hic site	es, year	ofis	solat	tion	and	num	ber o	f <i>L</i>	. maj	or i	sol	ates	inc	lude	d i	n th	e stu	dy.
-------	----	---------	----------	----------	------	-------	------	-----	-----	-------	------------	-------	------	-----	------	-----	------	-----	------	-------	-----

Site (governorate)	Recent isolates (2008-2012)	Historical isolates (1991–1992)
Metlaoui (Gafsa)	49	0
M'dhila (Gafsa)	17	0
Chrarda (Kairouan)	31	0
Mnara (Kairouan)	15	0
Oueled Haffouz (Sidi Bouzid)	24	0
Sidi Bouzid Centre (Sidi Bouzid)	0	35
Hania (Sidi Bouzid)	1	0
Total = 172 isolates		

doi:10.1371/journal.pntd.0004017.t001

automatic DNA sequencer. Fragment size was determined automatically using the GeneMapper 4.0 software.

Microsatellite data analysis

The FSTAT Version 2.9.3.2 software [23] updated from [24] was used to compute estimates and to test the significance of the various population genetic parameters. Genetic polymorphism was measured based on the allelic richness (A) and the Nei's unbiased estimate of genetic diversity within subsamples (Hs) [25]. The observed heterozygosity (Ho) and expected heterozygosity (He) were also calculated. The Wright's F statistics [26] were estimated with the Weir and Cockerham's method [27]: F_{IS} measures the relative inbreeding of individuals due to the local non-random union of gametes in each subpopulation, and $F_{\rm ST}$ measures the relative inbreeding in subpopulations attributable to the subdivision of the total population into subpopulations of limited size. Therefore, F_{ST} also measures the genetic differentiation between subpopulations. The significant departure from 0 of these parameters was tested by 10,000 randomization procedures with FSTAT. The genetic differentiation between historical and recent populations, between governorates (Gafsa, Kairouan and Sidi Bouzid) and between the two sub-populations (historical and recent focus) in the Kairouan governorate was explored. The significance of these estimates was confirmed by *p*-values ≤ 0.05 . The gene flow or migration rate between populations was also estimated using FSTAT, as $Nm = 1 - F_{ST}/4 F_{ST}$ [28]. A neighbor-joining tree [29], based on the Cavalli-Sforza and Edward's chord distances [30], was used to cluster the genotypes. Data were computed using the POPULATION software to build the distance matrix (version 1.2.28; CNRS, UPR9034, Langella, O.) and the tree was generated using FigTree, version 1.4.1 [31].

Finally, data were analyzed using a Bayesian model-based approach implemented in STRUCTURE, version 2.3.4 [32], to explore the structure of the *L. major* populations. STRUC-TURE uses Bayesian Monte-Carlo Markov Chain (MCMC) sampling to identify the optimal number of clusters K for a given multi-locus dataset, without requiring the identification of the population subunits a priori. The parameters used were the admixture model with the length of burn-in period of 200,000 iterations, followed by 200,000 MCMC repeats after burn-in. Based on the multi-locus genotype data, isolates were divided into K subpopulations with K ranging from 1 to 10 and ten independent runs were performed for each value of K. The K optimal value (i.e., the optimal number of clusters in the dataset) was calculated using STRUC-TURE HARVESTER, web version [33]. Two approaches were used to choose K. First, ΔK , which measures the second-order rate of change in the log likelihood of the data between successive values of K, was estimated [34]. Second, posterior probabilities for the values of K with the highest Ln P(X|K) were compared. STRUCTURE 2.3.4 was also used to identify migrants. In this case, prior population information was used in the USEPOPINFO option of STRUC-TURE. Populations defined according to geographic and temporal criteria (GF2008-2012, KR2008-2012 and SBZ2008-2012, SBZ1991-1992, MN2008-2012 and CH2008-2012) were used as prior population information for this test. Run conditions for this analysis were as mentioned above. As no information was available about migration, a range of migration rates was assigned (MIGPRIOR = 0.01, 0.05, 0.1), as a sensitivity test during the analysis.

Results

Analysis of L. major population structure in the different governorates

The data obtained from the ten polymorphic microsatellite markers [20] were used to assess and compare the genetic variability of isolates collected from patients in the three Tunisian governorates between 2008 and 2012. The Gafsa isolates (GF2008-2012; n = 66) included 48

Population (Area/number of isolates)	Descriptive statistics								
	A	Hs	Но	He	F _{IS}				
Historical population (Sidi Bouzid/35) SBZ1991-1992	2.4	0.129	0.1311	0.128	0.514				
Recent population (Sidi Bouzid /25) SBZ2008-2012	2	0.236	0.056	0.233	0.797				
Governorate 1 (Gafsa/66) GF2008-2012	3.2	0.321	0.059	0.319	0.816				
Governorate 2 (Kairouan/46) KR2008-2012	2.7	0.306	0.054	0.303	0.823				
Governorate 3 (Sidi Bouzid/25) SBZ2008-2012	2.0	0.236	0.048	0.233	0.797				
Historical focus (Mnara-Kairouan/15) MN2008-2012	2.2	0.278	0.033	0.271	0.880				
Emerging focus (Chrarda-Kairouan/31) CH2008-2012	2.3	0.316	0.064	0.314	0.796				
	Whole sam	ple (172)							

Table 2. Genetic diversity indices, estimated from microsatellite data (10 loci), for the 172 L. major isolates analyzed in this study.

L. major isolates were included in the different subpopulations: historical (SBZ1991-1992) and recent populations (SBZ2008-2012) from the Sidi Bouzid governorate; recent populations from Gafsa (GF2008-2012), Kairouan (KR2008-2012) and Sidi Bouzid (SBZ2008-2012); the (KR2008-2012) population was further divided in samples from Mnara (MN2008-2012, historical epidemic focus) and from Chrarda (CH2008-2012, emerging epidemic focus). A = allelic richness per population based on the standardized minimal sample size; Hs = gene diversity; Ho = observed heterozygosity; He = expected heterozygosity; $F_{IS} =$ inbreeding coefficient.

doi:10.1371/journal.pntd.0004017.t002

genotypes (genotypic diversity = 0.73), the Kairouan isolates (KR2008-2012; n = 46) 35 genotypes (genotypic diversity = 0.76) and the Sidi Bouzid isolates (SBZ2008-2012; n = 25) 24 different genotypes (genotypic diversity = 0.96). This finding shows the high *L. major* genotypic diversity in these three areas where ZCL is endemic. Unique genotypes and original microsatellite profiles were identified in 18 GF2008-2012 isolates, in four SBZ2008-2012 isolates and in 11 KR2008-2012 samples (S1 Table). Comparisons of the genetic diversity data for the three geographic groups revealed that intraspecific genetic diversity (H_s) was highest in the Gafsa and lowest in the Sidi Bouzid isolates (Table 2). Similarly, the allelic richness (*A*) and the mean observed heterozygosity (H_o) decreased progressively from Gafsa to Sidi Bouzid (Table 2).

The mean expected heterozygosity (H_e) values were much higher than the H_o values in all three populations (Table 2). These results were confirmed by the inbreeding coefficient (F_{IS}) values estimated for each locus and in each population, revealing a deficit in heterozygosity in all three populations (Table 2). The F_{ST} values, which are used as a measure of genetic differentiation between populations, were very low, but significantly different between the GF2008-2012 and SBZ2008-2012 isolates and between the GF2008-2012 and KR2008-2012 isolates. Conversely, the F_{ST} values were not significantly different between the KR2008-2012 and SBZ2008-2012 populations (Table 3). Accordingly, the highest migration rate (Nm) value was between the KR2008-2012 and SBZ2008-2012 populations, where as the Nm values were much

$rable 5$. Differentiation measures (r st), probabilities (r -value) and migration rate (r_m) between subpopulation	Table 3. D	ifferentiation measure	s (F _{ST}), probabilities	(P-value) and migration	on rate (N _m) betwee	en subpopulations.
--	------------	------------------------	-------------------------------------	-------------------------	----------------------------------	--------------------

Subpopulation (Area) (Number of isolates)	F _{ST}	P-value*	Nm (migrant/population)
Historical (Sidi Bouzid) (35) versus Recent (Sidi Bouzid) (25)	0.213	0.05	0.923
Governorate 1 (Gafsa) (66) versus Governorate 2 (Kairouan) (46)	0.034	0.05	7.03
Governorate 1 (Gafsa) (66) versus Governorate 3 (Sidi Bouzid) (25)	0.082	0.016	2.78
Governorate 2 (Kairouan) (46) versus Governorate 3 (Sidi Bouzid) (25)	0.008	0.083	31
Emerging focus (Chrarda-Kairouan) (31) versus Historical focus (Mnara-Kairouan) (17)	0.014	0.05	17.60

*Data were considered significant when *P*-value \leq 0.05.

doi:10.1371/journal.pntd.0004017.t003

lower between the GF2008-2012 and KR2008-2012 and between the GF2008-2012 and SBZ2008-2012 populations (Table 3).

The Bayesian model-based clustering analysis implemented in STRUCTURE indicated that our dataset (GF2008-2012, KR2008-2012 and SBZ2008-2012 isolates) could be organized in four (maximum L(K)) or two (maximum ΔK) clusters (K) (Fig 2A and 2B). For K = 2, 92% of L. major isolates from Gafsa, 10% from Kairouan and 38% from Sidi Bouzid were included in one of the inferred clusters and the remaining samples in the second one (Fig 2C.1). Based on the Q-matrix bar plots obtained for each isolate by calculating the posterior probabilities of belonging to each K cluster, the SBZ2008-2012 population showed a mixed membership to the inferred clusters (Fig 2C.1). For K = 4, the Q-matrix bar plots showed an increased separation of the L. major sample substructure. Although the most likely number of groups here suggested a total of four populations no strains were fully assigned to the fourth putative group (yellow group) (Fig 2C.2), suggesting that this was not a valid population for this set [34,35]. This "phantom" population suggesting a wider and deeper clinical sample collection may discover new diversity even in this small geographic area [36]. Based on the bar plots for the two assumptions, K = 2 seems to be the most probable partition for our data set. In agreement with the very low genetic differentiation between governorates, the analysis carried out with the STRUCTURE program did not divide the three populations according to their geographical origin. Nevertheless, most isolates from Gafsa and from Kairouan were grouped in cluster 1 and cluster 2, respectively, and the samples from Sidi Bouzid were distributed in the two clusters (38% in cluster 1 and 62% in cluster 2). Strains showing mixed membership were observed in each population, probably due to the low level of differentiation among populations. The occurrence of gene flow between GF2008-2012 and SBZ2008-2012, as well as between SBZ2008-2012 and KR2008-2012 can be clearly observed in Fig 2C.1. In the STRUCTURE assignment tests, we only reported the results for MIGPRIOR = 0.1, because migration appeared to occur frequently between governorates, thus the optimal MIGPRIOR value was likely to be the highest one [32]. Furthermore, when running the migration model at K = 3(equal to the number of predefined populations), ten samples were identified as migrants. Among these migrants, two GF2008-2012 samples were assigned to the KR2008-2012 population and one GF2008-2012 sample to the SBZ2008-2012 group. Among the KR2008-2012 samples, only two showed a posterior probability of having recently migrated from Sidi Bouzid. In the SBZ2008-2012 group, five samples were assigned to both Gafsa and Kairouan. These results show a recent connectivity between localities. According to the Q-values of the samples, the KR2008-2012 and SBZ2008-2012 populations seemed to be more inter-connected than the GF2008-2012 population with either SBZ2008-2012 or KR2008-2012.

The low genetic differentiation (F-statistical approach), the recent migration events (Bayesian analysis), the gene flow over the years (F-statistical approach) and the absence of clear separation (Bayesian analysis) between governorates were also confirmed by the finding that the phenetic tree did not highlight any subdivision of the strains according to their geographical origin. (<u>S1 Fig</u>).

Analysis of the *L. major* population structure in historical and recent isolates

The data of the ten microsatellite markers were used also to compare the 35 historical (SBZ1991-1992) and 25 recent (SBZ2008-2012) isolates from Sidi Bouzid governorate. Two of the microsatellites were monomorphic in the SBZ1991-1992 population, revealing 26 geno-types in the 35 isolates (0.74). Analysis of the SBZ2008-2012 isolates revealed three polymorphic microsatellites that generated 24 genotypes (0.96). The number of alleles per locus ranged



Fig 2. Estimated population structure of *L. major* in Tunisia as inferred by the STRUCTURE software on the basis of the data on 10 microsatellite markers obtained for 137 recent isolates from the Gafsa (GF2008-2012; n = 66), Kairouan (KR2008-2012; n = 46) and Sidi Bouzid (SBZ2008-2012; n = 66), Kairouan (KR2008-2012; n = 46) and Sidi Bouzid (SBZ2008-2012; n = 66), Kairouan (KR2008-2012; n = 46) and Sidi Bouzid (SBZ2008-2012; n = 66), Kairouan (KR2008-2012; n = 46) and Sidi Bouzid (SBZ2008-2012; n = 66), Kairouan (KR2008-2012; n = 46) and Sidi Bouzid (SBZ2008-2012; n = 66), Kairouan (KR2008-2012; n = 66), Kairouan (KR208-2012; n = 66), Kair

n = 25) governorates. (A) Plot of the mean posterior probability (LnP(D)) values per clusters (K), based on 10 replicates per K, generated by the STRUCTURE software, and (B) delta-K analysis of LnP(K). (C) STRUCTURE plots grouped by Q-matrix (estimated membership coefficient for each sample) showing the distribution of genetic variation (C.1) at K = 2 and (C.2) at K = 4. Each strain is represented by a vertical line, which is partitioned into the colored segments that represent the parasite estimated membership fractions in K. The same color indicates that the isolates belong to the same group. Different colors for the same isolate indicate the percentage of the genotype shared with each group. Gene flow between populations is indicated with arrows.

doi:10.1371/journal.pntd.0004017.g002

from 1 to 8, with an allelic richness (A) of 2.4 in the historical population and of 2 in the recent isolates (S1 Table). The pattern of unbiased gene diversity (Hs) increased over time (Table 2). Comparison of the mean H_e and H_o values showed a departure from the expected values in SBZ2008-2012 population, whereas both observed and expected heterozygosity were highly similar for SBZ1991-1992 population and no departure from expected values was detected for these strains (Table 2). Indeed, as reported in previous studies, H_o was much lower than H_e and the inbreeding coefficient $F_{\rm IS}$ revealed a strong heterozygote deficiency in both populations (Table 2). The $F_{\rm ST}$ value ($F_{\rm ST} = 0.213$) showed an important genetic differentiation between historical and recent isolates and the migration rate Nm was very low (0.923) (Table 3).

The Bayesian model-based clustering analysis indicated two clusters as the most probable genetic structure of these two populations. The first cluster included most of the historical samples (98.8%), which shared a common genetic background based on the genotyping results, and 19.3% of the recent isolates. The second cluster was mainly (80.7%) composed by recent isolates. STRUCTURE identified 0.42 and 4.82 misclassified samples for the historical and recent populations, respectively. However, as the migration model used different prior and modeling assumptions for identifying migrants, three SBZ2008-2012 isolates showed contrasting patterns of assignment, suggesting high genetic connectivity over time. The Neighbor Joining tree revealed no strict partition between the historical and recent populations (see Fig.3). The tree included five main clusters that corresponded to the two populations obtained with the STRUCTURE analysis: the two upper clusters matched the first STRUCTURE population, while the three remaining clusters corresponded to the second STRUCTURE population.

Analysis of the *L. major* population structure relative to the two ZCL emergence periods in Kairouan

The recent isolates from the Kairouan governorate (KR2008-2012, n = 46) came from the historical focus (1982) of Mnara (MN2008-2012, n = 15) and from the emerging focus (2005) of Chrarda (CH2008-2012, n = 31). Analysis of the genetic data of these two sub-populations showed that six of the ten tested microsatellite loci were polymorphic (4GTG, 39GTG, 45GTG, 1GC, 71AT and 1GACA) in both populations. The remaining four markers (27GTG, 36GTG, 28AT and 1CA) were monomorphic in one or the other population (S1 Table). The allelic richness (*A*) ranged from 2.2 (MN2008-2012 isolates) to 2.3 (CH2008-2012 isolates) (Table 2). *H*_s, *H*_o, *H*_e and *F*_{IS} were higher in the CH2008-2012 (emerging focus) than in the MN2008-2012 population. As in all the populations assessed in this study, the mean *H*_o values in the two sub-populations were extremely low compared to the *H*_e values. The heterozygosity deficit (measured by *F*_{IS}) was very high in both populations (Table 2).

The F_{ST} value showed that the genetic divergence in the two foci was very low, but significant (Table 3), indicating a genotype flow between populations. Quantification of the migration events in the two districts using F-statistical approaches, revealed 17 migrants per population. The Neighbor joining analysis confirmed that there was no genetic structuring between populations (S2 Fig). The posterior probabilities of belonging to a K cluster and the corresponding Δ K values calculated with the STRUCTURE software showed a peak at K = 2, indicating that two genetic clusters were the best solution for this dataset. The STRUCTURE



0.04



doi:10.1371/journal.pntd.0004017.g003

analysis did not detect any recent migrant over the last generations between the partitions defined by the software. The different migration rates observed with the two methods can be explained by the fact that STRUCTURE only identifies very recent migrants, whereas the F-

method may highlight migration events that occurred hundreds or thousands of years ago (Pritchard, personal communication).

Discussion

The Sidi Bouzid, Gafsa and Kairouan governorates are the areas where ZCL is most endemic in Tunisia. Despite the cyclic occurrence of outbreaks [8], the annual reported incidence was > 500 cases in Kairouan, and > 1000 cases in Gafsa and Sidi Bouzid in 2004 [37]. In this study, we carried out the first genetic substructure analysis of *L. major* populations in Tunisia based on genotype data to investigate the spatio-temporal dynamics and the evolution of this species in these three governorates where most human cases are recorded.

Spatial organization of L. major genetic diversity in Tunisia

Our findings revealed the presence of a decreasing genetic gradient from Gafsa to Kairouan and Sidi Bouzid between 2008 and 2012. According to the chronology of leishmaniasis emergence in Tunisia, Gafsa is the oldest and the most well-established focus of ZCL. The first outbreak was described in 1884 in El Guettar, in the southern part of the Gafsa governorate. Conversely, the first cases in the Kairouan and in Sidi Bouzid governorates were recorded only in 1982 and 1991, respectively. However, we do not know whether the chronology of disease emergence corresponds to the real spread of leishmaniasis. Furthermore, the governorate disease reports do not record the exact geographic origin of the infection.

Despite these limitations, this study shows the presence of some genetic differentiation between Gafsa and Sidi Bouzid and Gafsa and Kairouan, but not between Kairouan and Sidi Bouzid. Furthermore, we obtained a gradient of diversity that progressively decreases from Gafsa to Kairouan and then to Sidi Bouzid. These results are in agreement with the chronology of L. major outbreaks in these three governorates (see above) and do not follow a south-north geographical gradient. It is worth noting that this genetic pattern (i.e., population structuring according to the geography and to the ancientness of the focus) is classically observed for Leish*mania* species mainly because of the low dispersion capacity of sandflies and reservoirs [38]. Moreover, the STRUCTURE analysis could not cluster the three groups of recent isolates according to their provenance and showed more connectivity between Kairouan and Sidi Bouzid than between Gafsa and Sidi Bouzid. The analyses of the genotypic data, using the Bayesian model-based method and F-statistics, suggest that there had been and still there is a flow of genotypes between governorates, especially between Kairouan and Sidi Bouzid. According to the data published by Chargui et al. [18], L. infantum and L. major in Tunisia do not seem to follow the same genetic and evolutionary pattern. Indeed, although L. infantum seems to have spread from the north to the center of the country [16,39], more genetic diversity, particularly high heterozygosity, was found in the center than in the north.

Two hypotheses might explain the genetic differentiation and diversity patterns: 1) leishmaniasis spread first from Gafsa to Kairouan through human activities (economic activities or social development projects) and then from Kairouan to Sidi Bouzid; 2) parasites have evolved in different ways because of different ecosystems. Concerning the second hypothesis, rodent and sandfly ecology could influence the parasite transmission. Indeed, entomologic studies have demonstrated that in Tunisia sandfly populations vary in density and species composition according to the ecological conditions [40]. Furthermore, rodent populations (*P. obesus* and *M. shawi*) also show a different distribution in the different governorates linked to the food availability [11,41,42]. It is worth noting that, like for other *Leishmania* species (i.e., *L. guyanensis, L. braziliensis* and *L. donovani*), we found a strong deficit of heterozygosity in all our populations, in agreement with a recently described mixed-mating system of reproduction (clonality, endogamy, allogamy) [43,44,45,46,47]. Concerning *L. infantum* in Tunisia, high level of heterozygosity was observed in Kairouan with evidences of hybridization events [18]. These authors also found a deficit of heterozygosity, although they analyzed the MON-24 and MON-1 populations separately. The different genetic patterns and especially the higher heterozygosity observed in *L. infantum* compared to *L. major* could be explained by different proportions of the three reproduction mode of *Leishmania* parasites [43,44,45,46,47]. Nevertheless, the sample size in the work by Chargui et al. [18] was too small (27 strains) to do more extensive comparisons.

Temporal organization of L. major genetic diversity in Tunisia

First, we explored the evolution of the L. major population in isolates from Sidi Bouzid governorate between 1991–1992 and 2008–2012. The results show that the recent population is more diverse than the historical one with a significant genetic differentiation over time. In twenty years, L. major evolved with a change in allelic frequencies. The increase of genetic diversity reflects the accumulation of genetic changes overtime in this population. The continuous presence of some genotypes over the two decades and the grouping in the same cluster of both the recent and historical populations (Fig 2) strongly suggest that the recent population evolved from the historical one. This hypothesis is supported by the detection of migrants between the old and recent population. As ZCL incidence and Leishmania genetic diversity have been continuously increasing, we can assume that the Leishmania population has gradually adapted to the environment [7,8,48]. Furthermore, in the last twenty years, Tunisia, like the rest of the world, went through rapid ecosystem modifications. For example, the water project (construction of the Sid Saâd dam in the Nasrallah delegation, Kairouan governorate, in 1982) [49], the pest control program (destruction of rodent borrows and elimination of chenopods) and the development of agricultural projects around the city of Sidi Bouzid (Sidi Bouzid governorate, 1992) could have had an effect on temperature, humidity, soil and vegetation. These anthropic modifications might have disturbed the sandfly and rodent populations and thus impacted the evolution of the Leishmania population since the 1990's. Nevertheless, migration events from other regions cannot be excluded. The calculation of genetic differentiation between L. major population from Sidi Bouzid and those from North Africa [20], Central Asia [20], Middle East [20], Iran [19] and Pakistan [50] revealed considerable genetic differentiation (>0.55) with highly significant *p*-values (<0.05) (S2 Table).

To further understand the L. major temporal/spatial dynamics in Tunisia, we also analyzed recent isolates (2008–2012) from two districts within the Kairouan governorate: Mnara (historical focus) and Chrarda (emerging focus). Considering the long interval (23 years) between the human outbreaks in the two districts, we expected that the more recent focus would show a lower genetic diversity than the historical focus. However, the genetic data showed a slightly higher genetic diversity in the more recent focus (Chrarda), despite the low sample size. In parallel, the low differentiation suggests that these are not isolated populations. Indeed, the F-statistics and Bayesian methods estimated that the Mnara and Chrarda isolates are closely related, although the difference was sufficient to correctly assign most samples to their respective district. The migration analysis using both methods suggests the existence of historical migration events (F-statistics method), but not recent migration events (Bayesian method). These results support the hypothesis that the outbreak in Chrarda is the result of the spread of a population, rather than of a small set of genotypes, from Mnara several years ago. Based on the short distance between the Mnara and Chrarda districts (12 to 20 kilometers), the Leishmania population spread could be explained by human activities and also by the vector or reservoir dynamics. As sandflies are known to be bad flyers, Meriones shawi movements (the disease

reservoir) and the human economic and social exchanges could be the main sources of the emergence of *L. major* in Chrarda.

Conclusion

This study brings new insights into the spatial and temporal evolution of *L. major* in Tunisia. Over two decades, the *L. major* population evolved into a new, genetically differentiated population, probably better adapted to the environment. This could explain the increase of parasite transmission to humans and the higher incidence of ZCL in these areas over the last years [7,8,48].

To control the emergence of *L. major* in new areas in Tunisia, it is now essential to identify the routes of spread. Our findings suggest that the parasite population dynamics do not follow a vertical south-north gradient. Indeed, the disease seems to have spread from Gafsa to Kairouan and then to Sidi Bouzid. Human activities and/or the disease reservoir dynamics might explain this geographically non-gradual spread.

When a disease settles in a new area, it normally sources a subset of the original pathogen population and, as a consequence, genetic diversity should be reduced in the new population. Analysis of the Mnara and Chrarda isolates indicates that the two populations are similar with very low differentiation and historical migration events. This genetic similarity suggests the occurrence of high flow of genotypes between these neighboring populations that would be at the origin of the outbreak in Mnara. The relationships between environmental changes, human activities and reservoir systems have doubtlessly influenced the spread and the evolution of the Tunisian *L. major* populations as it is largely demonstrated for *Leishmania* species [2]. More work is needed to assess the influence of the movements and population structures of the rodent reservoirs and vectors on *L. major* evolution.

Supporting Information

S1 Fig. The unrooted neighbor-joining tree inferred from pair-wise Cavalli-Sforza and Edwards' chord distances computed from microsatellite data of 137 *Leishmania major* isolates (GF2008-2012, n = 66; green labels, KR2008-2012, n = 46; blue labels and SBZ2008-2012, n = 25, pink labels) does not show any subdivision of the strains according to their geographical origin.

(PDF)

S2 Fig. The Unrooted neighbor-joining tree inferred from genetic distances (Cavalli-Sforza and Edwards, 1967) computed from microsatellite data of 46 *L. major* strains (CH2008-2012 and MN2008-2012) shows no genetic structuring between the historical focus (MN2008-2012, n = 15; blue labels) and the emerging focus (CH2008-2012, n = 31; black labels).

(PDF)

S1 Table. Sample code, origin and multilocus microsatellite genotyping results of the *L. major* **isolates analyzed in this study.** The microsatellite profiles are designed by three-digit numbers separated by a fraction bar. Each three-digit number assigns the detected allele, coded by the microsatellite size in base pairs (only one size value for homozygous loci and two size values for heterozygous loci). (XLSX)

S2 Table. Differentiation measures (F_{ST}) between Tunisian *L. major* populations (historical and recent populations) and *L. major* isolates from Central Asia, Africa, Middle East, Iran

and Pakistan. (XLSX)

Acknowledgments

The authors want to thank the Regional Health Directorates of Sidi Bouzid, Kairouan, and Gafsa (Tunisia), the "Institut Pasteur Tunis" and the "Institut de Recherche pour le Développement" (IRD, Montpellier, France), the "Centre National de la Recherche Scientifique" (CNRS, France) for this cooperative and collaborative project. We acknowledge the field staff of the Department of Medical Biology, Institut Pasteur Tunis, for strain collections and the cryobank management; Adel Gharbi for ensuring strain and patients' data collection in accordance with the current Good Clinical Practice guidelines. Many thanks also to Dr. Francine Pratlong and Patrick Lami for providing samples from the French National Reference Center of Leishmania (CNRL, Montpellier, France) and Luc Abate (IRD, Montpellier, France) for help with the genotyping analysis. We are grateful to Pr. Jonathan K. Pritchard (Department of Biology and Genetics, Stanford University and Howard Hughes Medical Institute), Dr. Vikram E. Chhatre (University of Maryland, Center for Environmental Sciences), Dr. Dent Earl (Department of Biomolecular Engineering, Jack Baskin School of Engineering, University of California, Santa Cruz) for helpful comments and discussion. Great acknowledgments go to Pr. Koussay Dellagi from the Centre de Recherche et de Veille sur les Maladies Emergentes dans l'Océan Indien (CRVOI) for his useful comments on the manuscript. We thank Elisabetta Andermarcher for assistance in preparing and editing the manuscript.

Author Contributions

Conceived and designed the experiments: MHa MHi ALB. Performed the experiments: MHa AZ RY SC MHi ALB. Analyzed the data: MHa JB WG AT BC MHi ALB ABS. Contributed reagents/materials/analysis tools: MHa JB WG AZ RY SC MHi ALB ABS. Wrote the paper: MHa JB WG AT BC MHi ALB ABS.

References

- Déperet C, Boinet E (1884) Du bouton de Gafsa au champ de Sathomay. Arch Méd Pharm Mil 3: 296– 302.
- Ashford R (2000) The leishmaniases as emerging and reemerging zoonoses. International Journal for Parasitology 30: 1269–1281. PMID: <u>11113254</u>
- Bern C, Maguire JH, Alvar J (2008) Complexities of assessing the disease burden attributable to leishmaniasis. PLoS NeglTrop Dis 2: e313.
- Ben Salah A, Ben Messaoud N, Guedri E, Zaatour A, Ben Alaya N, et al. (2013) Topical Paromomycin with or without Gentamicin for Cutaneous Leishmaniasis. N Engl J Med 368: 524–532. doi: <u>10.1056/</u> <u>NEJMoa1202657</u> PMID: <u>23388004</u>
- Ben Salah A, Zakraoui H, Zaatour A, Ftaiti A, Zaafouri B, et al. (1995) A randomized, placebo-controlled trial in Tunisia treating cutaneous leishmaniasis with paromomycin ointment. Am J Trop Med Hyg 53: 162–166. PMID: <u>7677218</u>
- Ben Salah A, Buffet PA, Morizot G, Ben Massoud N, Zâatour A, et al. (2009) WR279,396, a third generation aminoglycoside ointment for the treatment of *Leishmania major* cutaneous leishmaniasis: a phase 2, randomized, double blind, placebo controlled study. PLoS Negl Trop Dis 3: e432. doi: <u>10.1371/journal.pntd.0000432</u> PMID: <u>19415122</u>
- 7. Salah AB, Kamarianakis Y, Chlif S, Alaya NB, Prastacos P (2007) Zoonotic cutaneous leishmaniasis in central Tunisia: spatio temporal dynamics. Int J Epidemiol 36: 991–1000. PMID: <u>17591639</u>
- Toumi A, Chlif S, Bettaieb J, Ben Alaya N, Boukthir A, et al. (2012) Temporal dynamics and impact of climate factors on the incidence of zoonotic cutaneous leishmaniasis in central Tunisia. PLoS Negl Trop Dis 6: e1633. doi: <u>10.1371/journal.pntd.0001633</u> PMID: <u>22563513</u>

- Ben Rachid M, Ben-Ismail R (1987) Current situation in regard to leishmaniasis in Tunisia. Research on Control Strategies for the Leishmaniases Proceedings of an International Workshop held in Ottawa June 1–4, Canada: IDRC-CIID, Manuscript report 184e.
- Anonyme Rapport annuel des Bulletins épidémiologiques de la Direction des soins de santé de base, Ministère de la Santé Publique-République Tunisienne Imprimerie officielle de la république Tunisienne 1980-2006.
- 11. Ben Ismail R, Ben Rachid MS (1989) Epidémiologie des leishmanioses en Tunisie. Mal Trop Trans 1: 73–80.
- Ben Ismail R, Gradoni L, Gramiccia M, Bettini S, Ben Rachid M, et al. (1986) Epidemic cutaneous leishmaniasis in Tunisia: biochemical characterization of parasites. Trans R Soc Trop Med Hyg 80: 669– 670. PMID: <u>3810806</u>
- 13. Rioux JP, Lanotte G, Petterb F, Derreure J, Akalay O, et al. (1986) The cutaneous leishmaniasis in the Western Mediterranean basin. From the enzymatic identification to the eco-epidemiological analysis. The example of three "focus" Tunisian, Moroccan and French, in: *Leishmania*. Taxonomy and phylogeny. Applications eco-epidemiological, Rioux JA (ed) IMEEE, Montpellier: 365–395.
- Ben Ismail R, Gramiccia M, Gradoni L, Helal H, Ben Rachid MS (1987) Isolation of *Leishmania major* from *Phlebotomus papatasi* in Tunisia. Transactions of the Royal Society of Tropical Medicine and Hygiene 81: 749. PMID: 3449994
- Ghawar W, Toumi A, Snoussi MA, Chlif S, Zaatour A, et al. (2011) *Leishmania major* infection among *Psammomys obesus* and *Meriones shawi*: reservoirs of zoonotic cutaneous leishmaniasis in Sidi Bouzid (central Tunisia). Vector Borne Zoonotic Dis 11: 1561–1568. doi: <u>10.1089/vbz.2011.0712</u> PMID: <u>21919726</u>
- Aoun K, Amri F, Chouihi E, Haouas N, Bedoui K, et al. (2008) Epidemiology of *Leishmania (L.) infantum*, *L. major and L. killicki* in Tunisia: results and analysis of the identification of 226 human and canine isolates. Bull Soc Pathol Exot 101: 323–328. PMID: 18956815
- Haouas N, Gorcii M, Chargui N, Aoun K, Bouratbine A, et al. (2007) Leishmaniasis in central and southern Tunisia: current geographical distribution of zymodemes. Parasite 14: 239–246. PMID: <u>17933302</u>
- Chargui N, Amro A, Haouas N, Schönian G, Babba H, et al. (2009) Population structure of Tunisian Leishmania infantum and evidence for the existence of hybrids and gene flow between genetically different populations. International Journal for Parasitology 39: 801–811. doi: <u>10.1016/j.ijpara.2008.11</u>. <u>016</u> PMID: <u>19211023</u>
- Mahnaz T, Al-Jawabreh A, Kuhls K, Schonian G (2011) Multilocus microsatellite typing shows three different genetic clusters of *Leishmania major* in Iran. Microbes Infect 13: 937–942. doi: <u>10.1016/j.micinf.</u> <u>2011.05.005</u> PMID: <u>21664984</u>
- Al-Jawabreh A, Diezmann S, Muller M, Wirth T, Schnur LF, et al. (2008) Identification of geographically distributed sub-populations of *Leishmania (Leishmania) major* by microsatellite analysis. BMC Evol Biol 8: 183. doi: 10.1186/1471-2148-8-183 PMID: 18577226
- Croan D, Morrison D, Ellis J (1997) Evolution of the genus *Leishmania* revealed by comparison of DNA and RNA polymerase gene sequences. Mol Biochem Parasitol 89(2): 149–159. PMID: <u>9364962</u>
- Rioux J, Lanotte G, Serres E, Pratlong F, Bastien P, et al. (1990) Taxonomyof *Leishmania*. Use of enzymes. Suggestions for a new classification. Ann Parasitol Hum Comp 65: 111–125. PMID: 2080829
- 23. Goudet J (2002) FSTAT: a program to estimate and test gene diversities and fixation indices. Version 2.9.3.2. Available at http://www.unil.ch/izea/softwares/fstat.html.
- Goudet J (1995) FSTAT (Version 1.2): A computer program to calculate F-statistics. J Hered 86: 485– 486.
- Nei M, Tajima F, Tateno Y (1983) Accuracy of estimated phylogenetic trees from molecular data. II. Gene frequency data. J Mol Evol 19: 153–170. PMID: 6571220
- Wright S (1978) Evolution and the genetics of natural populations, Variability within and among natural populations. University of Chicago Press, Chicago 4.
- Weir BS, Cockerham CC (1984) Estimating F-statistics for the analysis of population structure Evolution 38: 1358–1370.
- Souto CP, Premoli AC (2003) Genetic divergence among natural populations of Alstroemeria aurea D. Don: a dominant clonal herb of the under story in subalpine Nothofagus forests. Bot Bull Acad Sin 44: 329e336.
- Saitou N, Nei M (1987) The neighbor-joining method: a new method for reconstructing phylogenetic trees. Mol Biol Evol 4: 406–425. PMID: <u>3447015</u>
- Cavalli-Sforza LL, Edwards AW (1967) Phylogenetic analysis. Models and estimation procedures. Am J Hum Genet 19: 233–257. PMID: <u>6026583</u>

- 31. Rambaut A (2012) FigTree version 1.4.0. http://tree.bio.ed.ac.uk/software/figtree/.
- Pritchard JK, Stephens M, Donnelly P (2000) Inference of population structure using multilocus genotype data. Genetics 155(2): 945–959. PMID: <u>10835412</u>
- Earl DA, Von Holdt BM (2012) STRUCTURE HARVESTER: a website andprogram for visualizing STRUCTURE output and implementing the Evanno method. Conserv Genet Resour 4: 359–361. doi:310.1007/s12686-12011-19548-12687.
- Evanno G, Regnaut S, Goudet J (2005) Detecting the number of clusters of individuals using the software STRUCTURE: a simulation study. Mol Ecol 14: 2611–2620. PMID: <u>15969739</u>
- Hubisz MJ, Falush D, Stephens M, Pritchard JK (2009) Inferring weak population structure with the assistance of sample group information. Mol Ecol Resour 9: 1322–1332. doi: <u>10.1111/j.1755-0998.</u> <u>2009.02591.x</u> PMID: <u>21564903</u>
- Downing T, Stark O, Vanaerschot M, Imamura H, Sanders M, et al. (2012) Genome-wide SNP and microsatellite variation illuminate population-level epidemiology in the Leishmania donovani species complex. Infect Genet Evol 12: 149–159. doi: <u>10.1016/j.meegid.2011.11.005</u> PMID: <u>22119748</u>
- 37. Epidemiological Bulletins of the Direction of Primary Health care MoH, Tunisia (1980–2010).
- 38. Antoniou M, Gramiccia M, Molina R, Dvorak V, Volf P (2013) The role of indigenous phlebotomine sandflies and mammals in the spreading of leishmaniasis agents in the Mediterranean region. Euro Surveill 18: pii = 20540.
- Kallel K, Haouas N, Pratlong F, Kaouech E, Bel Hadj S, et al. (2008) Cutaneous leishmaniasis caused by *Leishmania infantum* MON 24 in Tunisia: extension of the focus to the center of the country. Bull Soc Pathol Exot 101: 30–31.
- Ghrab J, Rhim A, Bach-Hamba D, Chahed MK, Aoun K, et al. (2006) *Phlebotominae (Diptera: Psychodidae*) of human leishmaniosis sites in Tunisia. Parasite 13: 23–33. PMID: <u>16605064</u>
- Ben Ismail R, Ben Rachid MS, Gradoni L, Gramiccia M, Helal HB-H, D. (1987a) Zoonotic cutaneous leishmaniasis in Tunisia: study of the disease reservoir in the Douara area. Ann Soc Belg Med Trop 67: 335–343.
- 42. Fichet-Calvet E, Jomaa I, Ben Ismail R, Ashford RW (2003) Leishmania major infection in the fat sand rat Psammomys obesus in Tunisia: interaction of host and parasite populations. Ann Trop Med Parasitol 97: 593–603. PMID: 14511558
- Rougeron V, De Meeus T, Hide M, Waleckx E, Bermudez H, et al. (2009) Extreme inbreeding in *Leishmania braziliensis*. Proc Natl Acad Sci U S A 106: 10224–10229. doi: <u>10.1073/pnas.0904420106</u> PMID: <u>19497885</u>
- 44. Rougeron V, De Meeus T, Kako Ouraga S, Hide M, Bañuls AL (2010) "Everything you always wanted to know about sex (but were afraid to ask)" in *Leishmania* after two decades of laboratory and field analyses. PLoS Pathog 6: e1001004. doi: <u>10.1371/journal.ppat.1001004</u> PMID: <u>20808896</u>
- 45. Rougeron V, Bañuls AL, Carme B, Simon S, Couppié P, et al. (2011a) Reproductive strategies and population structure in *Leishmania*: substantial amount of sex in *Leishmania Viannia guyanensis*. Mol Ecol 20: 3116–3127.
- Rougeron V, Bañuls AL, Carme B, Simon S, Couppie P, et al. (2011b) Reproductive strategies and population structure in *Leishmania*: substantial amount of sex in *Leishmania Viannia guyanensis*. Mol Ecol 20: 3116–3127.
- Kuhls K, Cupolillo E, Silva S, Schweynoch C, Boité M, et al. (2013) Population structure and evidence for both clonality and recombination among Brazilian strains of the subgenus *Leishmania (Viannia)*. PLoS Negl Trop Dis 7: (10):e2490. doi: <u>10.1371/journal.pntd.0002490</u> PMID: <u>24205418</u>
- **48.** Chahed MK, Hsairi M, Bousnina A, Achour N, Bouratbine A, et al. (2002) Assessment of the cutaneous leishmaniasis surveillance system in Tunisia. Revue Tunisienne de la Santé Militaire: 40–47.
- 49. Ben Ammar R, Ben Ismail R, Helal H, Bach-Hamba D, Chaouch A, et al. (1984) Un nouveau foyer de leishmaniose cutanée de type rural dans la région de Sidi-Saad, Tunisie. Bull Soc Fr Parasitol 2: 9–12.
- Alam MZ, Bhutto AM, Soomro FR, Baloch JH, Nakao R, et al. (2014) Population genetics of Leishmania (Leishmania) major DNA isolated from cutaneous leishmaniasis patients in Pakistan based on multilocus microsatellite typing. Parasit Vectors 7: 332. doi: 10.1186/1756-3305-7-332 PMID: 25030377